Author's response to reviews

Title: Fully customized placement of orthodontic miniplates: A novel clinical technique

Authors:

Jan Hourfar (hourfar@t-online.de)
Georgios Kanavakis (gkanavak@gmail.com)
Peter Goellner (goellner@bluewin.ch)
Björn Ludwig (bludwig@kieferorthopaedie-mosel.de)

Version: 2 Date: 19 February 2014

Author's response to reviews:

1. Title page was added to the manuscript
2. Abstract was structured according to the guidelines
3. Competing interests of authors was added at the end of the manuscript
4. Authors contributions were also added at the end of the manuscript
5. This is not a research trial involving human subjects, therefore there was no need for ethical approval prior to preparing this manuscript.